

Temi di Discussione

(Working Papers)

Bank bonds: size, systemic relevance and the sovereign

by Andrea Zaghini





Temi di discussione

(Working papers)

Bank bonds: size, systemic relevance and the sovereign

by Andrea Zaghini

Number 966 - July 2014

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Giuseppe Ferrero, Pietro Tommasino, Margherita Bottero, Giuseppe Cappelletti, Francesco D'Amuri, Stefano Federico, Alessandro Notarpietro, Roberto Piazza, Concetta Rondinelli, Martino Tasso, Giordano Zevi. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

BANK BONDS: SIZE, SYSTEMIC RELEVANCE AND THE SOVEREIGN

by Andrea Zaghini*

Abstract

I analyze the risk premium on bank bonds at origination with special focus on the role of implicit and explicit public guarantees and the systemic relevance of issuing institutions. Looking at the asset swap spread on 5,500 bonds, I find that explicit guarantees and sovereign creditworthiness have a substantial effect on the risk premium. In addition, while large institutions still enjoy lower issuance costs linked to the TBTF framework, I find evidence of enhanced market discipline for systemically important banks which have faced an increased premium on bond placements since the onset of the financial crisis.

JEL Classification: G21, G18, G01. **Keywords**: too-big-to-fail, market discipline, sovereign guarantees, G-SIFIs.

Contents

1. Introduction	
2. Methodology and data	7
3. The cost of bonds	
4. Robustness	
5. Conclusion	
Appendix	
References	

^{*} Bank of Italy, Economic Outlook and Monetary Policy Directorate – Via Nazionale 91, 00184 Rome. Tel +39-06-47922994; Fax +39-06-47923720; E-mail: andrea.zaghini@bancaditalia.it.

1. Introduction¹

The financial crisis that originated in the US subprime mortgage market in the summer of 2007 has negatively affected banks' funding conditions for an extended period of time. A general overhauling of risk profiles at both the corporate and the sovereign levels occurred, especially in some peripheral euro-area countries. In addition, national and supranational measures aimed at supporting the financial system and extensive changes in prudential regulation have made some financial instruments more attractive to banks than others. The aim of this paper is to investigate the evolution of the cost of bond funding over a period that includes both the first wave of the global financial crisis and the subsequent euro-area sovereign debt market turmoil.

A distinctive feature of this paper is the analysis of the actual cost of funding to banks, namely the price at which bonds are sold on the primary market. By relying on the asset swap spread paid at origination on over 5,500 bonds, I analyse the role played by bank characteristics, issuance features and market sentiment. I focus on two issues: i) the role of the sovereign in providing both implicit support to the financial system and explicit guarantees on bank bonds and ii) the growing size and complexity of financial institutions. The second issue is related to the distortions of the too-big-to-fail safety net granted to very large banks (Mishkin 2006) and the uncertainty of the too-complex-to-price syndrome of systemic institutions (Haldane 2012). However, while the former issue suggests weakened market discipline, the latter implies enhanced market monitoring.

As for the role of the sovereign, there is empirical evidence that governments of strong creditworthiness provide an implicit guarantee to the domestic banking system (Sironi 2003; Gropp et al. 2011; Packer and Tarashev 2011; Ueda and Weder di Mauro 2013). This effect occurs via a higher credit rating assigned to those financial institutions which benefit from the implicit support. In particular, rating agencies often assign two different ratings to banks, which are usually referred to as "stand-alone" and "all-in" ratings. Both reflect the assessment of the probability of default by the bank, but only the

¹ The author would like to thank Piergiorgio Alessandri, Davide Avino, Olivier Bruno, Lorenzo Burlon, Andrea Cardillo, Cornelia Düwel, Giuseppe Grande, Aviram Levy, Sergio Masciantonio and Stefano Siviero for helpful discussions and useful suggestions. Editorial assistance by Alessandra Ferlesch and Christine Stone is gratefully acknowledged. The views expressed in the paper do not necessarily reflect those of the Bank of Italy.

latter includes the possibility of a public bail-out. According to this literature, the difference between the two ratings should represent the uplift (i.e. the implicit support) provided by the sovereign. In addition to the implicit support, explicit government guarantees on bank bonds were introduced after the collapse of Lehman Brothers by almost all advanced economies. While guarantees proved effective in restoring bank funding, they were also a source of distortions for the corporate bond market. Indeed, the pricing of such bonds was strongly clustered on a country basis, suggesting that in many instances "weak" banks from "strong" countries had access to cheaper funding than "strong" banks from "weak" countries (Levy and Zaghini 2011; Grande et al. 2011).

Regarding bank size, this work is related to a recent strand of the empirical literature that tries to distinguish between the issue of size *per se* of financial institutions, which eventually leads to the too-big-to-fail safety net benefits, and the systemic dimensions of banks (relative size, interconnectedness and complexity), which might make them too-difficult-to-save (Völz and Wedow 2011; Demirgüç-Kunt and Huizinga 2013; Bertay et al. 2013). While the too-big-to-fail problem — in connection with negative externalities and moral hazard — has long been identified (O'Hara and Shaw 1990), the systemic relevance of banks and their implications for financial stability have attracted the attention of academics and, in particular, regulators only since the eruption of the 2007 financial crisis (Acharya 2009; BCBS 2011; Bernanke 2012). From a global perspective, what makes a financial institution systemically relevant is not (only) the size of its balance sheet but (also) its magnitude relative to the domestic economy, its degree of substitutability, its cross-country activities and its business model (FSB 2011).

I find that, compared to AAA-rated governments, lower-rated sovereigns add a burden to the cost of debt issuance of domestic banks. This implicit negative support intensified in the euro-area sovereign debt crisis: I estimate that the absence of the backing of a AAA-rated government amounts, *ceteris paribus*, to an average increase of over 140 basis points in the funding cost. However, once I restrict the analysis to banks for which CDS spreads are priced — usually larger institutions that are more active in the bond market — I find that the risk premium more closely reflects the characteristics of each institution (soundness and creditworthiness), with the role of government somewhat reduced.

Furthermore, by distinguishing between banks' absolute size (total assets) and systemic relevance (being included in the Financial Stability Board's list of global

systemically important financial institutions, G-SIFIs), I find that financial investors were able to disentangle the two issues. My results show that the safety net benefits granted to too-big-to-fail institutions include lower funding costs on the primary bond market (the larger the bank, the lower the premium). Yet, I also find evidence of enhanced market discipline: since the onset of the global financial crisis, systemically important banks — which enjoyed a reduction of the spread before the crisis — pay, *ceteris paribus*, a larger premium on their bond issuance.

The rest of the paper is organised as follows. Section 2 describes the dataset and the econometric methodology; section 3 analyses the factors influencing bond yield at origination over the period 2006-2011; section 4 provides robustness checks and section 5 presents the conclusions.

2. Methodology and data

In this section I outline a panel regression of the premium paid by banks on the primary market to assess the determinants of the cost of bond funding over the six years from 2006 to 2011. Since the risk-free component of the funding cost is unavoidable for the issuer, I focus on the asset swap (ASW) spread, which is the difference between the actual bond yield and the fixed rate of the asset swap contract with similar characteristics. I do not follow the ASW spread evolution on the market after the day of issuance because, from the point of view of the issuer, the secondary market pricing of any debt security does not change the cost of already placed bonds. I also avoid the use of secondary market spreads because of the poor liquidity in the trading of some securities. In doing so I follow the methodology used in the early contributions by Morgan and Stiroh (2001) and Sironi (2003) for the banking sector, which have recently been applied to the debt issuance of non-financial corporations by Pianeselli and Zaghini (2014). Although this approach reduces the time-series dimension of the sample, it still leads to a large selection of bonds and issuing institutions.

The dataset contains all bonds with maturity at origination of at least 1 year for which the ASW spread at issuance is available from Thomson Reuters Datastream. In particular, the final sample includes 5514 bonds issued by 209 banks from 14 countries.² There are 879

²The full list of banks in the sample is reported in Table A1 of the Appendix.

bonds from banks headquartered in the US, 462 from the UK, 2173 from Germany and 2001 from other euro-area countries (Table 1).

30 25 2 15 5 10 20 Belgium -21.3 6.9 9.4 65.4 71.9 83.0 33 35 36 34 17 26 35 11 Cyprus -47.0 -31.8 250.4 35 11 Finland 63.2 58.3 26.3 91.6 6 1 2 1 1 2 4 5 10 France -29.7 -3.2 85.0 70.6 42.8 95.7 44 Germany -13.6 6.9 44.1 51.0 42.1 51.1 44 Gerece 69.3 134.7 268.0 236.8 107.6 401.6 19 Jreland -26.8 -7.0 68.7 343.3 325.6 15 13 2 1 32 Italy -22.2 6.2 73.8 98.0 95.3 173.3 99 36 41 17 36 63 110 33 33 34		2006	2007	2008	2009	2010	2011	Total
Belgium -21.3 6.9 9.4 65.4 71.9 83.0 3 35 36 34 17 26 35 11 Cyprus -47.0 -31.8 250.4 33 34 17 26 35 11 Finland 63.2 58.3 26.3 91.6 64 64 38 121 121 44 France -29.7 -3.2 85.0 70.6 42.8 95.7 44 Germany -13.6 6.9 44.1 51.0 42.1 51.1 44 Isis 153 92 601 616 555 22.1 Greece 69.3 134.7 268.0 236.8 107.6 401.6 19 Iteland -26.8 -7.0 68.7 349.3 325.6 15 131 26 110 33 92 100 21 601 33 100 33 39 53 29 118 </td <td>Austria</td> <td>-14.7</td> <td>-8.6</td> <td>31.8</td> <td>55.9</td> <td>36.3</td> <td>86.8</td> <td>14.9</td>	Austria	-14.7	-8.6	31.8	55.9	36.3	86.8	14.9
35 36 34 17 26 35 1 Cyprus -47.0 -31.8 250.4 35 36 I 2 I 1 2 1 35 36 36 36 37 37 38 35 36 37 37 37 37 37 37 37 37 37 37 37 37 37 38 121 121 44 45 37 44 45 37 44 45 44 45 44 45 44 45 44 46 38 121 121 44 44 46 38 121 121 44 46 38 121 121 44 46 46 38 121 121 44 46 46 38 121 121 44 47 46 47 47 36 61 13 2 17 43 47 48 47 48 47 48 47 48 47 48 47 48		30	25	2	15	5	10	87
Cyprus -47.0 -31.8 250.4 3.3 I 2 I 1 2 1 5.5 5.5 91.6 6.6 6.7 6.7 7.7 7.3.2 85.0 70.6 42.8 95.7 4.4 4 5 6.4 6.6 6.7 1.21 <td>Belgium</td> <td>-21.3</td> <td>6.9</td> <td>9.4</td> <td>65.4</td> <td>71.9</td> <td>83.0</td> <td>31.2</td>	Belgium	-21.3	6.9	9.4	65.4	71.9	83.0	31.2
I 2 I Finland 63.2 58.3 26.3 91.6 6 I 2 4 5 5 France -29.7 -3.2 85.0 70.6 42.8 95.7 44 Germany -13.6 6.9 44.1 51.0 42.1 51.1 44 Germany -13.6 6.9 44.1 51.0 42.1 51.1 44 Isis 155 153 92 601 616 555 24 Greece 69.3 134.7 268.0 236.8 107.6 401.6 19 Ireland -26.8 -7.0 68.7 349.3 325.6 15 Italy -22.2 6.2 73.8 98.0 95.3 173.3 99 36 41 17 36 63 110 33 93 33 29 133 55 Portugal -4.8 13.8 27.5 78.8 53.5 74.9 55 39 53 29		35	36	34	17	26	35	183
Finland 63.2 58.3 26.3 91.6 64.1 l 2 4 5 64.1 France -29.7 -3.2 85.0 70.6 42.8 95.7 44.1 55 59 46 38 121 121 44.1 Germany -13.6 6.9 44.1 51.0 42.1 51.1 44.1 155 153 92 601 616 555 24.1 Greece 69.3 134.7 268.0 236.8 107.6 401.6 19.5 5 7 5 13 2 1 35.6 16 15 Ireland -26.8 -7.0 68.7 349.3 325.6 15.1 16 15 6 10 21 66.10 38.100 36 41 17 36 63 110 38.100 Netherlands -4.8 13.8 27.5 78.8 53.5 74.9 39 53 29 118 155 131 55.131 9 9 8 16 5 4 55.131 55.90 20 41 28 31 55 90 22.133 101 54 27.2 98.4 98.1 160.8 110.0 8 16 5 4 55.1 90.1 23.55 90.12 101 120 187 98.140 140 144 90.144 49.144	Cyprus	-47.0	-31.8		250.4			34.9
I2451France -29.7 -3.2 85.0 70.6 42.8 95.7 44.1 55 59 46 38 121 121 44.1 Germany -13.6 6.9 44.1 51.0 42.1 51.1 44.1 155 153 92 601 616 555 24.1 Greee 69.3 134.7 268.0 236.8 107.6 401.6 19.5 Ireland -26.8 -7.0 68.7 349.3 325.6 15.1 44.1 16 15 6 10 21 60.1 63.3 110.0 38.10 Italy -22.2 6.2 73.8 98.0 95.3 173.3 94.10 36 41 17 36 $63.110.0$ $38.100.0$ $39.13.0$ Netherlands -4.8 13.8 27.5 78.8 53.5 74.9 $55.1.0$ 9 9 8 16 5 4 $55.1.0$ $55.90.0$ $22.3.091.3.0$ 9 9 8 16 5 4 $55.90.0$ $22.3.091.3.0$ $77.0.0.0.00.00.00.00.00.00.00.00.00.00.0$		1	2		1			4
France -29.7 -3.2 85.0 70.6 42.8 95.7 44.7 Germany -13.6 6.9 44.1 51.0 42.1 51.1 44.7 155 153 92 601 616 555 21.7 Greece 69.3 134.7 268.0 236.8 107.6 401.6 19.7 5 7 5 13 2 1 35.7 Ireland -26.8 -7.0 68.7 349.3 325.6 15.7 16 15 6 10 21 60.7 16 15 6 10 21 60.7 36 41 17 36 63 110 39 53 29 11.8 155 131 5 99 8 16 5 4 9 9 8 16 5 4 50 41 28 31 55 90 2 2 98.4 98.1 160.8 110.0 8 16 5 4 25 50 41 28 31 55 90 2 187 98.4 98.1 160.8 110.0 8 140 144 190 8	Finland			63.2	58.3	26.3	91.6	61.9
55 59 46 38 121 121 44 Germany -13.6 6.9 44.1 51.0 42.1 51.1 44 155 153 92 601 616 555 24 Greece 69.3 134.7 268.0 236.8 107.6 401.6 195 5 7 5 13 2 1 35 Ireland -26.8 -7.0 68.7 349.3 325.6 15 16 15 6 10 21 60 16 15 6 10 21 60 16 15 6 10 21 60 Netherlands -4.8 13.8 27.5 78.8 53.5 74.9 39 53 29 118 155 131 55 39 53 29 118 155 131 55 90 9 8 16 5 4 55 50 41 28 31 55 90 22 50 41 28 31 55 90 22 50 41 28 31 55 90 22 50 41 28 31 55 90 22 100 81 100 88 91 4 110 120 187 98 140 144 190 88				1	2	4	5	12
Germany -13.6 6.9 44.1 51.0 42.1 51.1 44.1 155 153 92 601 616 555 24 Greece 69.3 134.7 268.0 236.8 107.6 401.6 19 5 7 5 13 2 1 33 Ireland -26.8 -7.0 68.7 349.3 325.6 15 16 15 6 10 21 60 Italy -22.2 6.2 73.8 98.0 95.3 173.3 90 36 41 17 36 63 110 33 Netherlands -4.8 13.8 27.5 78.8 53.5 74.9 54 39 53 29 118 155 131 55 Portugal -48.7 -16.3 62.3 127.0 82.3 391.3 76 9 9 8 16 5 4 55 Spain -26.8 -3.6 95.8 122.5 163.6 238.5 122 50 41 28 31 55 90 2 United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 88 71 53 54 112 81 91 4 120 187 98 140 144 190 8	France	-29.7	-3.2	85.0	70.6	42.8	95.7	48.9
InstructionInstructi		55	59	46	38	121	121	440
InstantISSISS9260161655521Greece 69.3 134.7 268.0 236.8 107.6 401.6 19 575 13 21 35 Ireland -26.8 -7.0 68.7 349.3 325.6 15 161561021 66 161561021 66 18 -22.2 6.2 73.8 98.0 95.3 173.3 90 36 41 17 36 63 110 33 Netherlands -4.8 13.8 27.5 78.8 53.5 74.9 56 39 53 29 118 155 131 55 90 53 29 118 155 131 55 91 -48.7 -16.3 62.3 127.0 82.3 391.3 76 998 16 5 4 25 50 41 28 31 55 90 22 50 41 28 31 55 90 22 50 41 28 31 55 90 22 50 41 28 31 55 90 22 50 41 28 31 55 90 22 50 41 28 31 55 90 22 50 41 28 31 55 <	Germany	-13.6	6.9	44.1	51.0	42.1	51.1	40.5
Ireland 5 7 5 13 2 1 5 16 15 6 10 21 60 Italy -22.2 6.2 73.8 98.0 95.3 173.3 99.0 36 41 17 36 63 110 31 Netherlands -4.8 13.8 27.5 78.8 53.5 74.9 39 53 29 118 155 131 55 Portugal -48.7 -16.3 62.3 127.0 82.3 391.3 76 9 9 8 16 5 4 55 4 55 Spain -26.8 -3.6 95.8 122.5 163.6 238.5 122 50 41 28 31 55 90 2 United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 86 71 53 54 112 81 91 4 120 187 98 140 144 190 8		155	153	92	601	616	555	2172
Ireland -26.8 -7.0 68.7 349.3 325.6 15 16 15 6 10 21 60 Italy -22.2 6.2 73.8 98.0 95.3 173.3 90 36 41 17 36 63 110 3 Netherlands -4.8 13.8 27.5 78.8 53.5 74.9 54 39 53 29 118 155 131 57 Portugal -48.7 -16.3 62.3 127.0 82.3 391.3 76 9 9 8 16 5 4 55 Spain -26.8 -3.6 95.8 122.5 163.6 238.5 122 United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 86 71 53 54 112 81 91 4 120 187 98 140 144 190 88	Greece	69.3	134.7	268.0	236.8	107.6	401.6	191.7
Italy 16 15 6 10 21 60 Italy -22.2 6.2 73.8 98.0 95.3 173.3 99.3 36 41 17 36 63 110 33 Netherlands -4.8 13.8 27.5 78.8 53.5 74.9 54 39 53 29 118 155 131 55 Portugal -48.7 -16.3 62.3 127.0 82.3 391.3 76 9 9 8 16 5 4 55 Spain -26.8 -3.6 95.8 122.5 163.6 238.5 122 50 41 28 31 55 90 2 United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 86 71 53 54 112 81 91 4 120 187 98 140 144 190 8		5	7	5	13	2	1	33
Italy -22.2 6.2 73.8 98.0 95.3 173.3 99.3 36 41 17 36 63 110 33 Netherlands -4.8 13.8 27.5 78.8 53.5 74.9 39 53 29 118 155 131 55 Portugal -48.7 -16.3 62.3 127.0 82.3 391.3 76 9 9 8 16 5 4 55 Spain -26.8 -3.6 95.8 122.5 163.6 238.5 122 50 41 28 31 55 90 2 United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 86 71 53 54 112 81 91 4 120 187 98 140 144 190 8	Ireland	-26.8	-7.0	68.7	349.3	325.6		150.1
36 41 17 36 63 110 3 Netherlands -4.8 13.8 27.5 78.8 53.5 74.9 55.5 39 53 29 118 155 131 55.5 Portugal -48.7 -16.3 62.3 127.0 82.3 391.3 77.5 9 9 8 16 5 4 55.5 Spain -26.8 -3.6 95.8 122.5 163.6 238.5 122.5 United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 89.7 71 53 54 112 81 91 4 United States 9.1 55.1 105.0 126.7 194.7 197.0 11.1 120 187 98 140 144 190 88.7		16	15	6	10	21		68
36 41 17 36 63 110 3 Netherlands -4.8 13.8 27.5 78.8 53.5 74.9 55 39 53 29 118 155 131 55 Portugal -48.7 -16.3 62.3 127.0 82.3 391.3 77 9 9 8 16 5 4 25 Spain -26.8 -3.6 95.8 122.5 163.6 238.5 122 Spain -26.8 -3.6 95.8 122.5 163.6 238.5 122 United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 89 71 53 54 112 81 91 4 United States 9.1 55.1 105.0 126.7 194.7 197.0 111 120 187 98 140 144 190 88	Italy	-22.2	6.2	73.8	98.0	95.3	173.3	96.7
39 53 29 118 155 131 5 Portugal -48.7 -16.3 62.3 127.0 82.3 391.3 77 9 9 8 16 5 4 55 Spain -26.8 -3.6 95.8 122.5 163.6 238.5 12 50 41 28 31 55 90 2 United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 89 71 53 54 112 81 91 4 United States 9.1 55.1 105.0 126.7 194.7 197.0 111 120 187 98 140 144 190 8		36	41	17	36	63	110	303
Portugal -48.7 -16.3 62.3 127.0 82.3 391.3 7 9 9 8 16 5 4 5 Spain -26.8 -3.6 95.8 122.5 163.6 238.5 12 United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 88 United States 9.1 55.1 105.0 126.7 194.7 197.0 11 120 187 98 140 144 190 8	Netherlands	-4.8	13.8	27.5	78.8	53.5	74.9	54.8
9 9 8 16 5 4 5 Spain -26.8 -3.6 95.8 122.5 163.6 238.5 12 50 41 28 31 55 90 2 United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 80 71 53 54 112 81 91 4 United States 9.1 55.1 105.0 126.7 194.7 197.0 11 120 187 98 140 144 190 8		39	53	29	118	155	131	525
Spain -26.8 -3.6 95.8 122.5 163.6 238.5 12 50 41 28 31 55 90 2 United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 89 71 53 54 112 81 91 4 9.1 55.1 105.0 126.7 194.7 197.0 11 120 187 98 140 144 190 8	Portugal	-48.7	-16.3	62.3	127.0	82.3	391.3	76.9
50 41 28 31 55 90 2 United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 89 71 53 54 112 81 91 4 United States 9.1 55.1 105.0 126.7 194.7 197.0 111 120 187 98 140 144 190 8		9	9	8	16	5	4	51
United Kingdom 5.4 27.2 98.4 98.1 160.8 110.0 89 71 53 54 112 81 91 4 United States 9.1 55.1 105.0 126.7 194.7 197.0 111 120 187 98 140 144 190 8	Spain	-26.8	-3.6	95.8	122.5	163.6	238.5	120.2
71 53 54 112 81 91 4 United States 9.1 55.1 105.0 126.7 194.7 197.0 11 120 187 98 140 144 190 8	-	50	41	28	31	55	90	295
United States 9.1 55.1 105.0 126.7 194.7 197.0 11 120 187 98 140 144 190 8	United Kingdom	5.4	27.2	98.4	98.1	160.8	110.0	89.1
<u>120</u> 187 98 140 144 190 8	-	71	53	54	112	81	91	462
	United States	9.1	55.1	105.0	126.7	194.7	197.0	119.4
		120	187	98	140	144	190	879
Total -10.2 20.7 73.8 77.9 81.0 107.2 66	Total	-10.2	20.7	73.8	77.9	81.0	107.2	68.4

Sources: Dealogic and Thomson Reuters Datastream.

¹ Basis points. Number of bonds in italics.

The evolution of the risk premium over time reflects the two waves of the global financial crisis: starting from the tranquil year 2006, in which the banks in the sample paid an average of -10 basis points, the ASW spread began an upward trend that led to a peak of 107 basis points in 2011. However, the dynamics are extremely heterogeneous across countries. While many countries experienced a steady increase of the AWS spread over time (Italy, Spain and the US among them), some peaked in 2009, recovered the following year, but witnessed a new spike in 2011 (Austria, Finland and France). At the same time, Germany, after a relatively mild increase in the premium in 2008, levelled off between 40 and 50 basis points. The UK is the only country in the sample for which the average ASW premium at bond origination declined in 2011.

A major aspect of the global financial crisis is that it induced significant substitution effects among financial instruments, including those within the medium- to longer-term bond class. Indeed, given the widespread change in risk assessment, the increase in interest rate spreads and the drying-up of several sources of funding (as well as the consequences of the sovereign debt crisis just few years later), there was a significant adjustment in banks' funding sources (Cardillo and Zaghini 2012; ECB 2012). In addition, rescue plans by governments, monetary authorities and supranational organizations, together with changes in market regulations, have often amplified the substitution among different securities (CGFS 2011). For instance, the exacerbation of the financial crisis following the collapse of Lehman Brothers in September 2008 led the governments of many advanced economies to use unprecedented amounts of state aid to support the financial sector. Among the most valuable tools was the introduction of explicit government guarantees on bank fixed-income debt against the payment of a fee by the issuer. Government-guaranteed bonds quickly became a key source of bank funding (Panetta et al. 2009; Stolz and Wedow 2010). Notwithstanding the possible distortionary effects on bank risk taking and the unsolved quest for the "fair price" of public guarantees (Arping 2010; Gropp et al. 2011; Ejsing and Lemke 2011), financial institutions have made extensive use of such bonds: in the period from October 2008 to May 2010 close to 1400 guaranteed bonds were issued by approximately 200 banks from 17 countries, for an amount totalling more than €1 trillion (Lindh and Schich 2012).³

Table 2 reports country issuance by type of bond. They can be classified into four deal types: covered bonds, government-guaranteed bonds, senior bonds and subordinated bonds. While senior bonds are the bedrock of banks' issuance, the other kinds of bonds have also been used by a large number of institutions. Out of the 209 banks in the sample, 180 tapped the market with standard senior placements, 78 made use of covered bonds, 64 were able to place subordinated debt and 47 exploited the possibility of buying a public guarantee. While there is again strong heterogeneity across countries, the average cost of the four kinds of placement clearly reflects the risk of the type of deal, with subordinated debt paying a larger

³In addition to euro-area countries, the UK and the US, several other advanced economies introduced government guarantees on banks' debt (Australia, Denmark, New Zealand and South Korea, among others). Moreover, the issuance of bonds was also affected, at least in the euro area, by the two ECB covered bond purchase programmes (CBPP1 and CBPP2) implemented from the second half of 2009, under which the Eurosystem bought eligible covered bonds up to a nominal value of 60 billion and 640 billion respectively (Beirne et al. 2011).

spread than senior bonds (176 versus 77 basis points) and the two secured issuances being able to get a better price at origination: 45 basis points for covered bonds and 28 basis points for government-guaranteed issuances.

	СВ	GGB	SEN	SUB	Total
Austria	71.0	44.6	-1.1	62.2	14.9
	2	4	7	4	10
Belgium	23.3	33.8	46.0		31.2
-	1	1	3		3
Cyprus			34.9		34.9
			2		2
Finland	46.5		65.0		61.9
	1		1		2
France	42.8	-26.3	51.0	113.7	48.9
	3	1	18	6	18
Germany	26.7	156.6	55.6	154.8	40.5
	27	2	25	8	30
Greece	90.1	233.7	197.5	75.3	191.7
	1	1	3	1	4
Ireland		167.3	12.4	577.2	150.1
		3	3	2	4
Italy	90.7		88.5	186.5	96.7
	9		28	10	29
Netherlands	66.9	34.0	51.7	211.8	54.8
	2	5	7	5	10
Portugal	128.0	90.3	46.8	316.5	76.9
	4	3	6	1	7
Spain	169.6	72.1	37.8	293.6	120.2
	19	5	27	6	32
United Kingdom	80.2	15.5	98.4	147.1	89.1
	9	5	16	9	17
United States		6.4	137.7	94.9	119.4
		17	34	12	42
Total	44.7	28.2	77.1	176.1	68.4
	78	47	180	64	209

Sources: Dealogic and Thomson Reuters Datastream.

¹ Basis points. Number of issuing banks in italics.

The analysis of the determinants of the risk premium on bank bonds is based on two main sources of influence: the characteristics of the issuer and the characteristics of the bond itself. In addition, I also take into account that the market pricing can be directly and indirectly influenced by the sovereign's soundness. To disentangle the contribution of each group of variables I run the following regression by means of pooled OLS with time dummies:

$$spread_{i} = \alpha_{0} + \sum \alpha_{j} V_{i,j}^{issuer} + \sum \alpha_{k} V_{i,k}^{issue} + \sum \alpha_{l} V_{i,l}^{country} + \sum \alpha_{z} D_{z}^{time} + \varepsilon_{i}$$

where *spread* is the ASW spread at launch, V_j^{issuer} are the variables characterising the issuer (size, rating, CDS spread), V_k^{issue} are the bond features (volume, maturity, currency and rating), $V_i^{country}$ are the characteristics of the country of residence of the issuer parent (rating and geographical area) and D_z^{time} are (yearly) time dummies that take into account the market conditions at the time of the issuance.⁴ Table 3 reports the summary statistics of the main variables employed in the estimations (excluding dummy variables). All exogenous variables are taken at time *t* (the exact issuance day) with the exception of balance sheet data, which are lagged by one year.⁵

Ceteris paribus, I expect bonds with higher ratings to carry lower spreads. With regard to the size of the issue, institutions that are more creditworthy typically find it easier to place larger issues, but they may face higher costs (yields) to generate a sufficiently large demand for their placements. It follows that the relationship between the bond size and the spread is ambiguous. At the same time, banks that are more creditworthy usually find it easier to issue longer-term bonds, but this kind of bond tends to be coupled with a higher yield due to the longer redemption horizon. Again, the sign of the coefficient is a matter of empirical assessment.

⁴ The dataset includes information from several databases. In addition to Thomson Reuters Datastream, from which the ASW spread is sourced, banks' balance sheet dimensions and the numbers of employees come from SNL Financial and Bankscope, CDS spreads and government ratings come from Bloomberg, and the bond features (bond rating, maturity, volume, currency of denomination and type of deal) as well as the bank nationality and rating on the day of issuance come from DCM Analytics by Dealogic. Finally, the list of G-SIFIs is taken from FSB (2011, 2012).

⁵Regarding the dummy variables: the rating of the issuer and the rating of the bonds are the average of the ratings provided by Moody's, Fitch and Standard & Poor's linearised between 0 (C-) and 20 (AAA); the rating of the sovereign takes the value 0 for AAA-rated countries and 1 otherwise; the time dummies take the value 1 for each given year from 2006 to 2011 and 0 otherwise; the currency denomination dummy takes the value 1 for euro-denominated bonds and 0 otherwise; subordinated debt, covered bond and government-guaranteed bond dummies take the value 1 for each specific deal type and 0 otherwise; the sovereign debt crisis dummy takes the value 1 from 2010Q3 to 2011Q4 and 0 otherwise; the global financial crisis dummy takes the value 1 from 2007Q4 to 2011Q4 and 0 otherwise; and the pre-crisis dummy takes the value 1 in the period 2006Q1-2007Q3 and 0 otherwise.

As regards the size (log of total assets) of the banks, I expect a negative sign for the regression coefficient insofar as large banks are supposed to benefit from the implicit toobig-to-fail (TBTF) support of the government. In particular, the idea is that governments will not allow large financial institutions to go bankrupt when their failure would trigger significant disruptions in the domestic financial system. Thus, it is assumed that, because of TBTF support, investors expect the government to back the debts of these institutions should they face financial stress (Anginer and Warburton 2014; Santos 2014). This expectation is referred to as an implicit guarantee as there is no official commitment from the authorities.

able 3. Summary statistics						
	Observations	Mean	Median	Std. Dev.	Max	Min
ASW spread	5514	68.4	45.2	110	1119	-121
Total asset	5514	496237	273067	524741	2154650	18
Duration	5514	1990	1431.5	2242	36540	365
Volume	5514	400	100	649	15000	0.1
Bond rating	5514	15.09	15	2.57	20	2
Bank rating	5514	16.83	16	2.73	20	2
Bank CDS	1659	178.7	142.6	179.9	1569	5
Employees	5514	50136	21051	70321	312356	32

This table presents summary statistics. ASW spread is the difference between the bond yield and the fixed-leg rate of a swap contract with the same maturity (basis points). Total asset is the bank balance sheet value of all assets (millions of euros). Duration is the bond maturity at issuance (days). Bond Rating and Bank Rating are the average of the ratings provided by Moody's, Fitch and Standard & Poor's linearised between 0 (C-) and 20 (AAA). Bank CDS is the average of the daily credit default swap for 5-year contracts computed in the 15-day period before the bond issuance (basis points). Employees is the number of employees working for the bank.

A different and more recent concern relates to the systemic relevance of financial institutions. Indeed, the global financial crisis has highlighted the inadequacy of banking regulation, with too much focus on microprudential supervision and an almost neglected macroprudential policy (Borio 2011; Bernanke 2012). The collapse of Lehman Brothers in 2008 made clear how the turmoil following the failure of a single institution that was well-connected and with large exposures to many market segments may spill over to several

countries and pose a serious threat to global financial stability.⁶ The pricing of bonds of these large and complex institutions are subject to two different influences that carry opposing effects. On the one hand, the externalities associated with large institutions themselves perceived as TBTF may well be applied to their interconnectedness, complexity, lack of substitutability and global scope. On the other hand, the balance sheet of such institutions, linked to a business model that generally places greater emphasis on trading and capital market–related activities, may have become less transparent. After the eruption of the global financial crisis, the perceived risk of systemic institutions may well have changed, leading to the too-complex-to-price syndrome: they are becoming too complex to be managed in any effective way (Haldane 2012).

3. The cost of bonds

The first column of Table 4 shows the baseline regression. As expected, the bank characteristics suggest that the rating of the bank has a negative influence on the spread at launch: the better the rating, the lower the issuance cost.⁷ Among the issue features, the maturity of the bond at issuance is positively related to the cost, while the bond rating negatively affects the ASW spread. The decision to issue subordinated debt leads to an increase of 57 basis points with respect to senior bonds, while the coefficient of the covered bonds is not statistically different from zero. Another bond characteristic that is statistically significant is whether the issue is denominated in euros: bonds denominated in euros pay a spread that is 56 basis points cheaper than those in other currencies.

To take into account effects of the nationality of the issuer, I include a sovereign rating dummy variable in the baseline regression,⁸ with the idea that governments of strong creditworthiness provide an implicit guarantee to the whole domestic banking system. The

⁶The IMF/BIS/FSB (2009) Report was the first official publication dealing with the issue of financial institutions' systemic relevance; it provided the first "guidance for national authorities to assess the systemic importance of financial institutions, markets and instruments". The rule book by the Basel Committee on Banking Supervision (2011) contains instead the methodology to identify global systematically important financial institutions (G-SIFIs). The Financial Stability Board is in charge of publishing the list of G-SIFIs each year.

⁷The standard errors reported in the tables (from Table 4 to Table 7) are clustered by country of residence of the issuer. The statistical significance of the coefficients is not affected when clustering by type of deal.

⁸To take into account possible non-linearities in the relation, particularly due to the flight to quality phenomenon, the variable takes the value of 0 for AAA-rated countries and 1 otherwise. When running the robustness checks I use a linearised version of the variable.

variable turns out to be highly significant: banks from non-AAA states pay 80 basis points more to issue bonds than banks with AAA-rated sovereigns.

I then add two other variables in order to consider, first, the distinctive features of the government guaranteed issues and, second, the turbulence spilling over to the corporate bond market from the sovereign debt market from mid-2010 (Table 4, second column).⁹ As for the former, the support of the public scheme can be measured as an average reduction in the issuance premium of around 32 basis points. Regarding the spillover due to the sovereign debt crisis, the increase in the ASW spread at origination due to the crisis amounts to 33 basis points. However, these two coefficients do not consider the effect (negative or positive) of the creditworthiness of the sovereign in those particular contexts. I thus interact the two dummy variables with the sovereign rating variable. Column 3 of Table 4 shows that there are indeed significant differences between AAA-rated countries and the others. The explicit guarantee of a weak sovereign is worth 116 basis points less than the one from a top-rated sovereign. Given the large difference, this in turn suggests that it might well be the case that riskier banks (i.e. those with lower ratings) that are from sounder states could tap the bond market at a better price than sounder banks from weaker states. This confirms the finding in Levy and Zaghini (2011) and Grande et al. (2011) that in the guaranteed bank-debt market the security pricing strongly reflects the characteristics of the guarantor, while bank-specific and issue-specific factors play a minor role.

From the regression coefficients I can also compute the value of the government support during the period of turbulence in the sovereign debt market. The results show a significantly negative spillover from weak governments to bank funding costs. During the sovereign debt crisis the backing of a lower-than-AAA-rated sovereign amounts, *ceteris paribus*, to an increase of 143 basis points in the ASW spread paid at origination by banks headquartered in those countries.¹⁰

⁹ Note that the effect of all the other measures devised to support the financial system during the crisis — which do not show up explicitly as a bond feature — are incorporated in the rating of the bank.

¹⁰ Given that the "Sovereign debt crisis" variable is not significantly different from zero (Table 4, third column), the overall effect of a non-AAA-rated government is given by the sum of the "Weak sovereign rating" variable (33 bps) and the interaction between the "Weak sovereign rating" and the time dummy "Debt crisis" (110 bps). Also taking into account the "Sovereign debt crisis" variable would lead to a further deterioration of just 12 basis points.

Duration	0.011 ***	0.010 ***	0.010 ***
	0.002	0.002	0.001
Bond Rating	-8.378 ***	-7.820 ***	-7.172 ***
C	2.965	2.850	2.337
Subordinated debt	57.01 *	66.25 *	65.50 *
	39.278	39.830	39.679
Covered Bonds	3.667	7.425	4.02
	18.035	16.993	14.821
Bank Rating	-7.754 **	-7.554 **	-7.640 **
-	3.448	3.438	3.470
lssuance in euros	-56.17 ***	-56.32 ***	-52.92 ***
	12.494	11.895	10.361
Weak Sovereign Rating	79.88 **	75.06 **	33.49 **
	29.994	28.904	26.744
Government Guarantee		-32.13 *	-44.53 *
		19.071	11.277
Sovereign Debt Crisis		33.35 *	12.36
		18.592	8.661
SovRat*GovGuarant			116.3 *
			71.376
SovRat*DebtCrisis			109.8 ***
			19.040

¹ Dependent variable: ASW spread; included observations: 5514; clustered standard errors by country; symbols *, ** and *** denote statistical significance at the 10%, 5% and 1% levels, respectively.

In order to have a more detailed picture of bank bond issuance, and as a robustness check, I now restrict the sample to banks for which a CDS is priced — usually larger institutions that are more active on the debt market. The number of bonds is reduced to 1659, issued by 142 banks. Even though the sample reduction is sizeable, I introduce into the empirical investigation an important quantitative variable describing the market perception of the soundness and creditworthiness of each institution.

The first column of Table 5 shows that the CDS coefficient is highly significant and displays the expected sign: a deterioration in the perceived soundness of the bank (an

increase in the CDS) leads to an increase in the cost of funding.¹¹ For the period of the financial crisis, the cost of debt issuance for these institutions is also cheaper when accompanied by the public guarantee and more expensive during the sovereign debt crisis (second column).

0.014 ***	0.013 ***	0.014 ***
0.002	0.002	0.002
-9.388 ***	-9.411 ***	-9.715 ***
1.389	1.388	1.474
131.3 ***	136.9 ***	139.8 *
29.662	29.542	29.984
19.54 *	19.71 *	15.04
10.262	10.197	10.130
-12.17 ***	-11.72 ***	-11.43 ***
2.393	2.369	2.356
-38.20 ***	-38.13 ***	-37.49 ***
6.510	6.528	6.553
73.80 ***	70.41 ***	47.38 ***
7.624	7.576	10.121
0.061 ***	0.050 ***	0.053 ***
0.014	0.014	0.014
	-39.85 ***	-45.46 ***
	12.355	12.457
	17.93 *	-0.072
	7.605	6.926
		17.73
		27.778
		62.76 ***
		16.959
	-9.388 *** 1.389 131.3 *** 29.662 19.54 * 10.262 -12.17 *** 2.393 -38.20 *** 6.510 73.80 *** 7.624 0.061 ***	-9.388 *** -9.411 *** 1.389 1.388 131.3 *** 136.9 *** 29.662 29.542 19.54 * 19.71 * 10.262 10.197 -12.17 *** -11.72 *** 2.393 2.369 -38.20 *** -38.13 *** 6.510 6.528 73.80 *** 70.41 *** 7.624 7.576 0.061 *** 0.050 *** 0.014 0.014 -39.85 *** 12.355 17.93 *

¹ Dependent variable: ASW spread; included observations: 1659; clustered standard errors by country; symbols *, ** and *** denote statistical significance at the 10%, 5% and 1% levels, respectively.

However, when assessing the creditworthiness of the sovereign as guarantor and during the crisis (third column), two circumstances stand out. First, the fact that a lower-rated government is backing the debt issuance does not have an additional (negative) effect on the cost of guaranteed bank bonds. Second, for the whole sample of banks, the sovereign debt crisis seems to affect only the issuers headquartered in the lower rated countries —

¹¹ Note that CDS spreads price not only the default risk of the bank but also the liquidity premium on the outstanding debt of that institution (Bongaerts et al. 2011; Badaoui et al. 2013).

those more exposed to the crisis. These findings suggest that the market is attaching more importance to the characteristics of the bank (part of the risk being captured by the CDS). When I compute the difference between the guaranteed issuance of banks in lower-rated countries and that from banks in AAA-rated countries, the spread is only 47 basis points (the "Weak sovereign rating" dummy). At the same time, the weakness of sovereigns significantly spilled over to the home banking systems during the sovereign debt crisis: the difference between top-rated and lower-rated countries amounted to 110 basis points.

As a further step in the analysis I check for a relationship between banks' dimension/systemic weight and the premium paid on bonds by introducing in the empirical framework both the size of the balance sheet (measured by the log of total assets) and a variable identifying the banks' systemic relevance (being in the FSB's 2011 list of the 29 G-SIFIs).

The first column of Table 6 shows that bank size has a negative coefficient, consistent with the too-big-to-fail hypothesis, confirming that a larger size tends to induce lower funding costs (Acharya et al. 2013; Santos 2014). At the same time, the coefficient for the systemic relevance of financial institutions is not significantly different from zero, suggesting that systemic relevance might not be an additional issue with respect to the TBTF framework. However, when taking into account the likely non-linearity of the relationship, due to the fact that systemic relevance is a more recent concern, different evidence emerges. The systemic relevance coefficient shows significantly different behaviour in the period before the disruption of the US subprime market (2006Q1-2007Q3) from the period during the global financial crisis (2007Q4- 2011Q4). Before the crisis, the market assessment of systemically important banks was benevolent: there was a large discount of 62 basis points in the risk premium associated with bond issuance of G-SIFIs (Table 6, second column). It is very likely that the implications of the systemic relevance of financial institutions were reinforcing the TBTF argument. Yet, financial agents' attitude towards those institutions changed direction during the crisis: debt issuance began to occur at a higher premium (37 basis points), suggesting the emergence of active market monitoring. Thus, these results support the hypothesis that market participants are now aware of the new framework in which G-SIFIs operate and opt for a more cautious approach when dealing with their debt. The outbreak of the crisis and the demise of Lehman Brothers in particular may have acted as a wake-up call for investors. These results are confirmed when estimating a richer

	Pool	ed	Fixed Ef	ffects
	(a)		(b)	(c)
Duration	0.012 ***	0.012 ***	0.012 ***	0.010 ***
	0.002	0.002	0.002	0.003
Bond rating	-8.005 ***	-6.831 ***	-7.039 ***	-8.416 ***
_	2.283	2.244	1.961	1.404
Bank rating	-7.371 **	-8.083 **	-8.458 **	-7.962 *
	3.464	3.638	3.510	5.315
Issuance in euros	-27.24 **	-25.62 **	-26.60 **	-39.08 **
	10.175	9.110	4.436	17.040
Sovereign debt crisis	39.93 *	28.10 *	28.12 *	28.64 **
-	22.154	18.642	18.020	9.859
Total assets	-15.23 **	-17.52 ***	-15.51 ***	-16.37 **
	7.245	6.340	2.617	8.834
G-SIFI	8.312			
	16.740			
G-SIFI*Pre-Global Crisis		-61.82 ***	-66.10 ***	-69.19 **
		15.453	10.937	17.225
G-SIFI*Post-Global Crisis		37.10 **	32.61 ***	38.11 **
		16.935	11.161	15.738
R-squared	0.234	0.235	0.342	0.339

econometric framework. Column 3 and column 4 of Table 6 show the panel estimations when I introduce the country fixed effects and the type of deal fixed effects.

1 Dependent variable: ASW spread; included observations: 5514; (a) clustered standard errors by country; (b) fixed effects by country; (c) fixed effects by type of deal; symbols *, ** and *** denote statistical significance at 10%, 5% and 1% levels, respectively.

These results are in line with the recent empirical literature. Enhanced market discipline for banks of systemic relevance is also found by Bertay et al. (2013), who analyse the cost and growth of deposits as market indicators of bank funding costs. In addition, by looking at the CDS market Völz and Wedow (2011) find a negative coefficient on the size variable and a positive coefficient on the square of the same variable, which is taken as the index of the systemic relevance of banks or, in their words, as the indicator that banks have become too-big-to-rescue.

4. Robustness

As robustness checks of these results I use several alternative definitions of G-SIFIs: the updated list provided by FSB (2012), which identifies 28 G-SIFIs, the combination of the two lists FSB (2011) and FSB (2012), which leads to 31 G-SIFIs, and the ranking provided by Masciantonio (2013), which replicates the FSB methodology using publicly available market data (27 G-SIFIs). Given the high correlation between the rating of the bond and the rating of the issuer, I drop one of the two alternately in the regressions. I also employ a variable for the sovereign rating constructed with the same linearisation applied to the rating of the issuer and the bond instead of the AAA dummy. Finally, I rely on different time windows for the definition of "financial crisis" and "sovereign debt crisis". My findings about the sovereign influence and the difference between size and systemic relevance are not affected.

Bond rating	0.002	0.002	
Bond rating		0.002	0.002
	-4.481	-4.467	-6.122 ***
	4.365	4.391	4.589
Bank rating	-19.33 ***	-19.34 ***	-20.18 **
	3.412	3.350	4.447
Issuance in euros	-40.73 ***	-41.34 ***	-39.14 **
	10.540	10.715	5.557
Weak sovereign rating	172.2 ***	125.0 *	
	51.625	75.681	
Sovereign debt crisis	19.23 *	8.84 **	
	12.631	4.923	
SovRat*DebtCrisis		77.39 *	
		45.913	
Total assets			-14.78 ***
			9.467
G-SIFI*Pre-Global Crisi	S		-70.49 ***
			11.459
G-SIFI*Post-Global Cris	is		26.57 ***
			8.728

¹ Dependent variable: ASW spread; included observations: 3587; fixed effects by country; symbols *, ** and *** denote statistical significance at 10%, 5% and 1% levels, respectively.

I run a further test for the sub-sample of senior unsecured bonds, which is a set of much more homogeneous bonds that forms the base of banks' long-term funding. Regressions for the 3587 senior bonds confirm the main findings of the paper. First, banks that are headquartered in countries with a rating lower than AAA face, *ceteris paribus*, much higher costs than peers located in top-rated economies — a difference of 172 basis points (Table 7, column 1). Moreover, the strain of the sovereign debt crisis is almost entirely felt in lower-rated countries (an increase in the ASW spread of 86 basis points), whereas the additional cost paid by banks in AAA-rated economies is just 9 basis points (Table 7, column 2). Finally, the evolution of market sentiment with respect to the issuance activity of G-SIFIs is also confirmed: before the global financial crisis, banks that were labelled systemically important benefitted from a sizeable discount (70 basis points), whereas during the crisis G-SIFIs underwent enhanced market discipline, which translates into an increase of 27 basis points in the premium paid at origination (Table 7, column 3).

5. Conclusion

The paper provides an assessment of the determinants of the premium paid on bond issuance by banks in the US, the euro area and the UK. I focus on a period (2006-2011) that includes the whole of the global financial crisis starting in the summer of 2007, which evolved into a painful sovereign debt crisis in several euro-area countries. The crisis has caused a deterioration in banks' funding conditions, leading in some cases to the drying-up of funding sources, the impairment of market segments and significant substitution effects among financial instruments. In addition, starting from mid-2010, concerns about the sustainability of public finances in several euro-area countries led to a deterioration in perceived sovereign creditworthiness. In parallel with the worsening of funding conditions in the home country and the related sovereign downgrades by rating agencies, many banks suffered the same fate, with increasing CDS spreads and widespread downgrades by several notches, further impairing banks' funding conditions.

In order to disentangle the factors affecting the cost of bond issuance I constructed an empirical investigation of the risk premium at origination on 5,500 bonds. I found that the backing of a AAA-rated sovereign provides an important implicit support to the home banking system, while weaker governments increase the funding cost of banks. I estimate that during the most acute phase of the sovereign debt crisis the absence of a AAA-rated

government implicit support amounts, *ceteris paribus*, to an increase of 143 basis points in the ASW spread paid at launch by domestic banks. Furthermore, in line with the recent literature on government-guaranteed bank bonds, I observe that the security pricing of explicitly guaranteed debt largely reflects the soundness of the sovereign. Yet, when looking at banks having a CDS — usually institutions which tap bond markets more regularly — I find that premia required by investors are more closely related to the characteristics of each bank, with the sovereign role partially downsized.

All in all, these findings suggest that there is a direct linkage between sovereign creditworthiness and (domestic) banks' funding cost, and that this linkage is particularly strong in crisis periods. This in turn implies that rating agencies play an important role in the market pricing process. Indeed, when they assign a sovereign rating, they directly influence the sovereign debt market and indirectly the corporate bond market. Thus not only the grade of the rating but also the timing of the decision can have significant implications for the funding choice of the domestic banking system (Cole and Cooley 2014).

In addition, by focusing on balance sheet dimension and degree of complexity of banks, I investigate whether the size and the systemic relevance of financial institutions matter for the pricing of bonds at origination. Indeed, the substantial involvement of large and complex financial institutions in the international spreading of the financial crisis has already prompted a reassessment of the rules under which they operate, aimed at making the failure of G-SIFIs less likely and the impact of their bankruptcy less widespread and costly (BCBS 2011). I find that this process makes a difference in banks' funding conditions and that financial investors are able to disentangle the two issues of size and systemic relevance. On the one hand, my results suggest that the size of the bank is negatively associated with the cost of bond financing, thus expanding the list of the of too-big-to-fail benefits to a reduced premium paid on the primary debt market. On the other hand, I find evidence of enhanced market discipline of systemically important banks in the late phase of the global financial crisis, with G-SIFIs facing an increased premium on their debt issuance. The latter finding points to a change in the market perception of the risk of G-SIFIs and to a shift towards a closer scrutiny of large and complex financial institutions. In particular, bondholders may consider it more likely than before that they will become involved (bailedin) in the case of a managed resolution. However, it might well be that once the cross-border supervision of G-SIFIs is fully in place, with capital adequacy ratio surcharges and crossborder coordinated recovery and resolution plans, financial markets will assess the risk of such institutions differently.

Appendix

Austria (10)	Caisse Nationale des Caisses d'Epargne et de Prevoyance SA - CNCE	Deutsche Genossenschafts-Hypothekenbank AG
Ceska SporitelnaAS	Compagnie Financiere du Credit Mutuel	Deutsche Hypothekenbank AG
Erste Group Bank AG	Credit Agricole	Deutsche Postbank AG
Hypo Alpe-Adria-Bank International AG	Credit Foncier de France SA - CFF	Deutsche Schiffsbank AG
Hypo Tirol Bank AG	Credit Industriel et Commercial SA - CIC	DVB Bank AG
KAAG	DEXIA Credit Local	DZ Bank AG
Kommunalkredit	Emporiki Group Finance plc	Eurohypo AG
Oesterreichische Volksbanken AG	Findomestic Banca SpA	HSH Nordbank AG
Raiffeisen Bank International AG	Fortis Bank SA/NV	HYPO REAL ESTATE
Raiffeisen Zentralbank Oesterreich AG - RZB	Groupe Credit Mutuel CEE	Landesbank Baden-Wuerttemberg - LBBW
Vorarlberger Landes- und Hypothekenbank AG	Klepierre financing	Landesbank Berlin AG
Belgium (3)	NATIXIS SA	Landesbank Hessen-Thueringen Girozentrale - Helaba
DEXIA Bank	SGA Societe Generale Acceptance NV	Landesbank Rheinland-Pfalz Girozentrale - LRP
IIB Capital plc	Societe Generale	Landesbank Saar
KBC Bank NV	UkrSibbank AKIB	Muenchener Hypothekenbank eG
Cyprus (2)	Germany (30)	Norddeutsche Landesbank Girozentrale - NORD/LB
Bank of Cyprus Public Co Ltd	Aareal Bank AG	Sparkasse KoelnBonn
Marfin Popular Bank Public Co	Bank Forum OAO	Westdeutsche ImmobilienBank AG
Finland (2)	Bayerische Landesbank	WestLB AG
OP Mortgage Bank	Berlin-Hannoversche Hypothekenbank AG	WL Bank AG Westfaelische Landschaft Bodenkreditbank
Pohjola Bank plc	Bremer Landesbank Kreditanstalt Oldenburg Girozentrale	Wuestenrot Bank AG Pfandbriefbank
France (18)	Commerzbank AG	Greece (4)
Banque Federative du Credit Mutuel - BFCM	Corealcredit Bank AG	EFG Hellas plc
BNP Paribas	DAPO Bank	FinansBank AS
Caisse Centrale du Credit Immobilier de France - 3CIF	Deutsche Bank AG	National Bank of Greece SA

Piraeus Group Finance plc	Centro Leasing Banca SpA	Banco Espirito Santo
Ireland (4)	CIB Bank Ltd	Banif
Allied Irish Banks plc	Credito Emiliano SpA	BES
Anglo Irish Bank	Credito Valtellinese Scarl - Creval	Caixa Economica Montepio Geral
Bank of Ireland	Intesa Sanpaolo SpA	Caixa Geral Depo
Irish Nationwide Building Society	Mediobanca	Spain (32)
Italy (29)	Sanpaolo IMI SpA	Abbey National Treasury Services pk
Banca Carige SpA	Ukrsotsbank PJSC	Alliance & Leicester plc
Banca delle Marche SpA	UniCredit Bank	Banco de Sabadell SA
Banca IMI SpA	Unione di Banche Italiane Scpa - UBI Banca	Banco de Valencia SA
Banca Italease SpA	Veneto Banca Holding ScpA	Banco Pastor SA
Banca Lombarda e Piemontese SpA	Netherlands (10)	Banco Popular Espanol SA
Banca Monte dei Paschi di Siena SpA - MPS	ABN AMRO Bank NV	Banesto
Banca Popolare dell'Alto Adige - Sudtiroler Volksbank	Achmea Hypo Bank	Bankinter
Banca Popolare dell'Emilia Romagna Scarl	Fortis Bank	BBVA
Banca Popolare dell'Etruria e del Lazio Scarl	Friesland Bank NV	Caixa de Ahorros de Vigo Ourense e Pontevedra - Caixanova
Banca Popolare di Cividale Scarl	ING Bank NV	Caixa d'Estalvis de Catalunya
Banca Popolare di Milano Scarl	Leaseplan	Caixa d'Estalvis de Girona
Banca Popolare di Vicenza Scarl	NIBC Bank NV	Caixa d'Estalvis de Terrassa
Banca Popolare Italiana Scarl	Rabobank Nederland	Caixa d'Estalvis del Penedes
Banco Popolare Scarl	SNS Bank NV	Caixa Girona
Bayerische Hypo- und Vereinsbank AG - HVB Group	Tango Finance Corp	Caja de Ahorros de Castilla la Mancha - CCM
Capitalia SpA	Portugal (7)	Caja de Ahorros de Murcia - Caja Murcia
Cassa di Risparmio di Bolzano SpA (Suedtiroler Sparkasse AG)	Banco BPI	Caja de Ahorros de Valencia Castellon y Alicante
Cassa di Risparmio di Ferrara SpA	Banco Comercial Portugues	Caja de Ahorros del Mediterraneo - CAM

Table A1 Banks in the sample by parent nationality (continued)

Table A1 Banks in the sample by parent nationali	ty (continued)		
Caja de Ahorros Municipal de Burgos	Principality Building Society	HSBC Bank plc	
Caja de Ahorros y Monte de Piedad de Avila - Caja de Avila	Royal Bank of Scotland	Jefferies Group Inc	
Caja de Ahorros y Monte de Piedad de Madrid - Caja Madrid	Skipton Building Society	JP Morgan Chase & Co	
Caja de Ahorros y Pensiones de Barcelona - La Caixa	Standard Chartered plc	KEYCORP	
Caja Espana de Inversiones Salamanca y Soria Caja de Ahorros	Ulster Bank Finance plc	M&T Bank Corp	
Caja Granada	Yorkshire Building Society	Mellon Funding Corp	
Cajamar	United States (42)	METLIFE INC	
Catalunya Caixa - Caixa d'Estalvis de Catalunya Tarragona & Manresa	Associated Banc-Corp	Morgan Stanley	
Ibercaja	Bank of America Corp	National City Bank	
Kutxa	Bank of New York Mellon Corp	Northern Trust Corp	
Newcastle Building Society	BB&T Corp	NY Community Bank	
Santander	Capital One Financial Corp	PNC Funding Corp	
Univaja	Centauri Corp	Regions Bank	
United Kingdom (17)	CIT Group Inc	State Street Corp	
ABSA Bank Ltd	Citigroup Inc	SunTrust Bank	
Bank of Scotland	City National Corp	SVB Financial Group	
Barclays Bank plc	Comerica Bank	US Bancorp	
Bradford & Bingley plc	Countrywide Financial Corp	USI Holdings Corp	
Coventry Building Society	Fifth Third Bancorp	Wachovia Bank NA	
Dunfermline Building Society	First Midwest Bancorp Inc	Washington Mutual Inc	
Hang Seng Bank Ltd	First Niagara Group Inc	Wells Fargo & Co	
HBOS	Fulton Corp	Western Alliance Bancorp	
HSBC Bank plc	Genelral Eelectric CAP CRP	Wilmington Trust Corp	
Leeds Building Society	GMAC INC	Zions Bancorp	
Lloyds TSB Bank plc	Goldman Sachs Group Inc		

References

Acharya V.V. (2009), "A theory of systemic risk and design of prudential bank regulation", *Journal of Financial Stability*, Vol.5, No.3, pp.224-255.

Acharya, V.V., Anginer D. and Warburton A.J. (2013), 'The End of Market Discipline? Investor Expectations of Implicit State Guarantees' New York University Stern School of Business Working Paper.

Anginer D. and Warburton A.J. (2014), "The Chrysler Effect: The Impact of Government Intervention on Borrowing Costs." *Journal of Banking and Finance*, Vol.40, No.1, pp.62-79.

Arping S. (2010), "The pricing of bank debt guarantees", *Economic Letters*, Vol.108, No.2, pp.119-121.

BCBS, Basel Commitee on Banking Supervision (2011), "Global systemically important banks: assessment methodology and the additional loss absorbency requirement", Bank for International settlements.

Badaoui S., Cathcart L, El-Jahel L. (2013), "Do sovereign credit default swaps represent a clean measure of sovereign default risk? A factor model approach", *Journal of Banking and Finance*, Vol. 37, pp.2392-2407.

Beirne J., Dalitz L., Ejsing J., Grothe M., Manganelli S., Monar F., Sahel B., Sušec M., Tapking J. and Vong T., (2011), "The impact of the Eurosystem's covered bond purchase programme on the primary and secondary markets" ECB Occasional Paper 122.

Bernanke, B. (2012), "Fostering Financial Stability", Board of Governor of the Federal Reserve System, Speech No.2012-0409.

Bertay A.C, Demirgüç-Kunt A. and Huizinga H. (2013), "Do we need big banks? Evidence on performance, strategy and market discipline", *Journal of Financial Intermediation*. Vol.22, No.4, pp.532-558.

Bongaerts D., De Jong F. and Driessen J. (2011), "Derivative Pricing with Liquidity Risk: Theory and Evidence from the Credit Default Swap Market", *Journal of Finance*, Vol. 66, pp. 203-240.

Borio, C. (2011), "Rediscovering the Macroeconomic Roots of Financial Stability Policy: Journeys, Challenges and a Way Forward", BIS Working Paper No.354.

Cardillo A. and Zaghini A. (2012), "Recent trends in long-term bank funding conditions", Questioni di Economia e Finanza (Occasional Papers) No.137, Bank of Italy.

CGFS, Committee on the Global Financial System (2011), "The impact of sovereign credit risk on bank funding conditions", CGFS Papers, No. 43.

Cole H. and Cooley T.F. (2014), 'Rating Agencies', NBER Working Paper No. 19972.

Demirgüç-Kunt A. and Huizinga H. (2013), "Are banks too big to fail or too big to save? International evidence from equity prices and CDS spreads", *Journal of Banking and Finance*, Vol.37, pp.875–894.

ECB, European Central Bank (2012), *Changes in bank financing patterns*, Frankfurt am Main.

Ejsing J. and Lemke W. (2011), "The Janus-Headed Salvation: Sovereign and Bank Credit Risk Premia during 2008-2009", *Economics Letters*, Vol.110, No.1, pp.28-31.

FSB, Financial Stability Board (2011), Policy measures to address systematically important financial institutions, Basel.

FSB, Financial Stability Board (2012), Update of group of global systemically important banks (G-SIBs), Basel.

Grande G., Levy A., Panetta F. and Zaghini A. (2011), "Public guarantees on bank bonds: effectiveness and distortions", *OECD Journal: Financial Market Trends*, Vol. 2011, No.2.

Gropp R., Hakenes H. and Schnabel I. (2011), "Competition, Risk-shifting, and Public Bailout Policies", *Review of Financial Studies*, Vol.24, No.6, pp.2084-2120.

Haldane, A. (2012), "We should go further unbundling banks", Speech No.605, Bank of England.

IMF/BIS/FSB, International Monetary Fund, Bank for International Settlements, Financial Stability Board (2009), "Guidance to Assess the Systemic Importance of Financial Institutions, Markets and Instruments: Initial Considerations", Report to the G-20 Finance Ministers and Central Bank Governors.

Levy A. and Zaghini A. (2011), "The pricing of government-guaranteed bank bonds", *Banks and Bank Systems*, Vol. 6, No. 3, pp. 16-24.

Lindh S. and Schich S. (2012), "Implicit guarantees for Bank Debt: Where Do We Stand?", *OECD Journal: Financial Market Trends*, Vol. 2012, No. 1.

Masciantonio, S. (2013), "Identifying and Tracking Systematically Important Financial Institutions (SIFIs) with public data", Bank of Italy Occasional Paper, No. 204.

Mishkin, F. (2006), "How big a problem is too big to fail?", *Journal of Economic Literature*, Vol.44, No.4, pp.988-1004.

Morgan, D.P. and Stiroh K.J. (2001). "Market Discipline of Banks: the Asset Test", *Journal of Financial Services Research*, Vol. 20, pp.195-208

O'Hara M. and Shaw W. (1990), "Deposit insurance and wealth effects: the value of being 'too big to fail'", *Journal of Finance*, Vol. 45, pp.1587–1600.

Packer F. and N.A. Tarashev (2011) "Rating Methodologies for Banks", *BIS Quarterly Review*, June 2011.

Panetta F., Faeh T., Grande G., Ho C., King M., Levy A., Signoretti F., Taboga M. and Zaghini A. (2009), "An assessment of financial sector rescue programmes", BIS Papers, No.48.

Pianeselli D. and Zaghini A. (2014), "The cost of firms' debt financing and the global financial crisis', *Finance Research Letters*, Vol.11, No.2, pp.74-83.

Santos J. (2014), "Evidence from the bond market on banks' too-big-to-fail subsidy", *Economic Policy Review*, Vol.20, No.2.

Sironi, A. (2003), "Testing for market discipline in the European banking industry: evidence from subordinated debt issues", *Journal of Money, Credit and Banking*, Vol.35, No.3, pp.443-472.

Stolz S.M. and Wedow M. (2010), "Extraordinary Measures in Extraordinary Times, Public Measures in Support of the Financial Sector in the EU and the United States", ECB Occasional Paper No.117.

Ueda, K. and Weder di Mauro B. (2013), "Quantifying structural subsidy values for systemically important financial institutions", *Journal of Banking and Finance*, Vol.37, pp.3830-3842.

Völz M. and Wedow M. (2011), "Market discipline and too-big-to-fail in the CDS market: Does banks' size reduce market discipline?", *Journal of Empirical Finance*, Vol.18, pp.195-210.

RECENTLY PUBLISHED "TEMI" (*)

- N. 940 Heterogeneous firms and credit frictions: a general equilibrium analysis of market entry decisions, by Sara Formai (November 2013).
- N. 941 The trend-cycle decomposition of output and the Phillips curve: Bayesian estimates for Italy, by Fabio Busetti and Michele Caivano (November 2013).
- N. 942 Supply tightening or lack of demand? An analysis of credit developments during the Lehman Brothers and the sovereign debt crises, by Paolo Del Giovane, Andrea Nobili and Federico Maria Signoretti (November 2013).
- N. 943 Sovereign risk, monetary policy and fiscal multipliers: a structural model-based assessment, by Alberto Locarno, Alessandro Notarpietro and Massimiliano Pisani (November 2013).
- N. 944 *Calibrating the Italian smile with time-varying volatility and heavy-tailed models*, by Michele Leonardo Bianchi, Frank J. Fabozzi and Svetlozar T. Rachev (January 2014).
- N. 945 *Simple banking: profitability and the yield curve*, by Piergiorgio Alessandri and Benjamin Nelson (January 2014).
- N. 946 Information acquisition and learning from prices over the business cycle, by Taneli Mäkinen and Björn Ohl (January 2014).
- N. 947 *Time series models with an EGB2 conditional distribution*, by Michele Caivano and Andrew Harvey (January 2014).
- N. 948 Trade and finance: is there more than just 'trade finance'? Evidence from matched bank-firm data, by Silvia Del Prete and Stefano Federico (January 2014).
- N. 949 *Natural disasters, growth and institutions: a tale of two earthquakes*, by Guglielmo Barone and Sauro Mocetti (January 2014).
- N. 950 *The cost of firms' debt financing and the global financial crisis*, by Daniele Pianeselli and Andrea Zaghini (February 2014).
- N. 951 On bank credit risk: systemic or bank-specific? Evidence from the US and UK, by Junye Li and Gabriele Zinna (February 2014).
- N. 952 School cheating and social capital, by Marco Paccagnella and Paolo Sestito (February 2014).
- N. 953 The impact of local minimum wages on employment: evidence from Italy in the 1950s, by Guido de Blasio and Samuele Poy (March 2014).
- N. 954 *Two EGARCH models and one fat tail*, by Michele Caivano and Andrew Harvey (March 2014).
- N. 955 *My parents taught me. Evidence on the family transmission of values*, by Giuseppe Albanese, Guido de Blasio and Paolo Sestito (March 2014).
- N. 956 Political selection in the skilled city, by Antonio Accetturo (March 2014).
- N. 957 *Calibrating the Italian smile with time-varying volatility and heavy-tailed models*, by Michele Leonardo Bianchi (April 2014).
- N. 958 *The intergenerational transmission of reading: is a good example the best sermon?*, by Anna Laura Mancini, Chiara Monfardini and Silvia Pasqua (April 2014).
- N. 959 A tale of an unwanted outcome: transfers and local endowments of trust and cooperation, by Antonio Accetturo, Guido de Blasio and Lorenzo Ricci (April 2014).
- N. 960 The impact of R&D subsidies on firm innovation, by Raffaello Bronzini and Paolo Piselli (April 2014).
- N. 961 *Public expenditure distribution, voting, and growth*, by Lorenzo Burlon (April 2014).
- N. 962 *Cooperative R&D networks among firms and public research institutions*, by Marco Marinucci (June 2014).
- N. 963 *Technical progress, retraining cost and early retirement*, by Lorenzo Burlon and Montserrat Vilalta-Bufí (June 2014).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- S. DI ADDARIO, *Job search in thick markets*, Journal of Urban Economics, v. 69, 3, pp. 303-318, **TD No.** 605 (December 2006).
- F. SCHIVARDI and E. VIVIANO, *Entry barriers in retail trade*, Economic Journal, v. 121, 551, pp. 145-170, **TD** No. 616 (February 2007).
- G. FERRERO, A. NOBILI and P. PASSIGLIA, Assessing excess liquidity in the Euro Area: the role of sectoral distribution of money, Applied Economics, v. 43, 23, pp. 3213-3230, **TD No. 627 (April 2007).**
- P. E. MISTRULLI, Assessing financial contagion in the interbank market: maximum entropy versus observed interbank lending patterns, Journal of Banking & Finance, v. 35, 5, pp. 1114-1127, TD No. 641 (September 2007).
- E. CIAPANNA, *Directed matching with endogenous markov probability: clients or competitors?*, The RAND Journal of Economics, v. 42, 1, pp. 92-120, **TD No. 665 (April 2008).**
- M. BUGAMELLI and F. PATERNÒ, *Output growth volatility and remittances*, Economica, v. 78, 311, pp. 480-500, **TD No. 673 (June 2008).**
- V. DI GIACINTO e M. PAGNINI, Local and global agglomeration patterns: two econometrics-based indicators, Regional Science and Urban Economics, v. 41, 3, pp. 266-280, TD No. 674 (June 2008).
- G. BARONE and F. CINGANO, Service regulation and growth: evidence from OECD countries, Economic Journal, v. 121, 555, pp. 931-957, TD No. 675 (June 2008).
- P. SESTITO and E. VIVIANO, *Reservation wages: explaining some puzzling regional patterns*, Labour, v. 25, 1, pp. 63-88, **TD No. 696 (December 2008).**
- R. GIORDANO and P. TOMMASINO, *What determines debt intolerance? The role of political and monetary institutions*, European Journal of Political Economy, v. 27, 3, pp. 471-484, **TD No. 700 (January 2009).**
- P. ANGELINI, A. NOBILI e C. PICILLO, *The interbank market after August 2007: What has changed, and why?*, Journal of Money, Credit and Banking, v. 43, 5, pp. 923-958, **TD No. 731 (October 2009).**
- G. BARONE and S. MOCETTI, *Tax morale and public spending inefficiency*, International Tax and Public Finance, v. 18, 6, pp. 724-49, **TD No. 732 (November 2009).**
- L. FORNI, A. GERALI and M. PISANI, *The Macroeconomics of Fiscal Consolidation in a Monetary Union:* the Case of Italy, in Luigi Paganetto (ed.), Recovery after the crisis. Perspectives and policies, VDM Verlag Dr. Muller, **TD No. 747 (March 2010).**
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap changes before and during the subprime financial turmoil, in Barbara L. Campos and Janet P. Wilkins (eds.), The Financial Crisis: Issues in Business, Finance and Global Economics, New York, Nova Science Publishers, Inc., TD No. 749 (March 2010).
- A. LEVY and A. ZAGHINI, *The pricing of government guaranteed bank bonds*, Banks and Bank Systems, v. 6, 3, pp. 16-24, **TD No. 753 (March 2010).**
- G. BARONE, R. FELICI and M. PAGNINI, *Switching costs in local credit markets,* International Journal of Industrial Organization, v. 29, 6, pp. 694-704, **TD No. 760 (June 2010).**
- G. BARBIERI, C. ROSSETTI e P. SESTITO, The determinants of teacher mobility: evidence using Italian teachers' transfer applications, Economics of Education Review, v. 30, 6, pp. 1430-1444, TD No. 761 (marzo 2010).
- G. GRANDE and I. VISCO, *A public guarantee of a minimum return to defined contribution pension scheme members*, The Journal of Risk, v. 13, 3, pp. 3-43, **TD No. 762 (June 2010).**
- P. DEL GIOVANE, G. ERAMO and A. NOBILI, Disentangling demand and supply in credit developments: a survey-based analysis for Italy, Journal of Banking and Finance, v. 35, 10, pp. 2719-2732, TD No. 764 (June 2010).
- G. BARONE and S. MOCETTI, With a little help from abroad: the effect of low-skilled immigration on the female labour supply, Labour Economics, v. 18, 5, pp. 664-675, **TD No. 766 (July 2010).**
- S. FEDERICO and A. FELETTIGH, *Measuring the price elasticity of import demand in the destination markets of italian exports,* Economia e Politica Industriale, v. 38, 1, pp. 127-162, **TD No. 776 (October 2010).**
- S. MAGRI and R. PICO, *The rise of risk-based pricing of mortgage interest rates in Italy*, Journal of Banking and Finance, v. 35, 5, pp. 1277-1290, **TD No. 778 (October 2010).**

- M. TABOGA, Under/over-valuation of the stock market and cyclically adjusted earnings, International Finance, v. 14, 1, pp. 135-164, **TD No. 780 (December 2010).**
- S. NERI, *Housing, consumption and monetary policy: how different are the U.S. and the Euro area?*, Journal of Banking and Finance, v.35, 11, pp. 3019-3041, **TD No. 807 (April 2011).**
- V. CUCINIELLO, *The welfare effect of foreign monetary conservatism with non-atomistic wage setters*, Journal of Money, Credit and Banking, v. 43, 8, pp. 1719-1734, **TD No. 810 (June 2011).**
- A. CALZA and A. ZAGHINI, welfare costs of inflation and the circulation of US currency abroad, The B.E. Journal of Macroeconomics, v. 11, 1, Art. 12, **TD No. 812 (June 2011).**
- I. FAIELLA, *La spesa energetica delle famiglie italiane*, Energia, v. 32, 4, pp. 40-46, **TD No. 822 (September 2011).**
- D. DEPALO and R. GIORDANO, *The public-private pay gap: a robust quantile approach*, Giornale degli Economisti e Annali di Economia, v. 70, 1, pp. 25-64, **TD No. 824 (September 2011).**
- R. DE BONIS and A. SILVESTRINI, *The effects of financial and real wealth on consumption: new evidence from* OECD countries, Applied Financial Economics, v. 21, 5, pp. 409–425, **TD No. 837 (November 2011).**
- F. CAPRIOLI, P. RIZZA and P. TOMMASINO, *Optimal fiscal policy when agents fear government default*, Revue Economique, v. 62, 6, pp. 1031-1043, **TD No. 859 (March 2012).**

2012

- F. CINGANO and A. ROSOLIA, *People I know: job search and social networks*, Journal of Labor Economics, v. 30, 2, pp. 291-332, **TD No. 600 (September 2006).**
- G. GOBBI and R. ZIZZA, Does the underground economy hold back financial deepening? Evidence from the italian credit market, Economia Marche, Review of Regional Studies, v. 31, 1, pp. 1-29, TD No. 646 (November 2006).
- S. MOCETTI, *Educational choices and the selection process before and after compulsory school*, Education Economics, v. 20, 2, pp. 189-209, **TD No. 691 (September 2008).**
- P. PINOTTI, M. BIANCHI and P. BUONANNO, *Do immigrants cause crime?*, Journal of the European Economic Association, v. 10, 6, pp. 1318–1347, **TD No. 698 (December 2008).**
- M. PERICOLI and M. TABOGA, *Bond risk premia, macroeconomic fundamentals and the exchange rate*, International Review of Economics and Finance, v. 22, 1, pp. 42-65, **TD No. 699 (January 2009).**
- F. LIPPI and A. NOBILI, *Oil and the macroeconomy: a quantitative structural analysis,* Journal of European Economic Association, v. 10, 5, pp. 1059-1083, **TD No. 704 (March 2009).**
- G. ASCARI and T. ROPELE, *Disinflation in a DSGE perspective: sacrifice ratio or welfare gain ratio?*, Journal of Economic Dynamics and Control, v. 36, 2, pp. 169-182, **TD No. 736 (January 2010).**
- S. FEDERICO, *Headquarter intensity and the choice between outsourcing versus integration at home or abroad*, Industrial and Corporate Chang, v. 21, 6, pp. 1337-1358, **TD No. 742 (February 2010).**
- I. BUONO and G. LALANNE, *The effect of the Uruguay Round on the intensive and extensive margins of trade*, Journal of International Economics, v. 86, 2, pp. 269-283, **TD No. 743 (February 2010).**
- A. BRANDOLINI, S. MAGRI and T. M SMEEDING, Asset-based measurement of poverty, In D. J. Besharov and K. A. Couch (eds), Counting the Poor: New Thinking About European Poverty Measures and Lessons for the United States, Oxford and New York: Oxford University Press, TD No. 755 (March 2010).
- S. GOMES, P. JACQUINOT and M. PISANI, The EAGLE. A model for policy analysis of macroeconomic interdependence in the euro area, Economic Modelling, v. 29, 5, pp. 1686-1714, TD No. 770 (July 2010).
- A. ACCETTURO and G. DE BLASIO, Policies for local development: an evaluation of Italy's "Patti Territoriali", Regional Science and Urban Economics, v. 42, 1-2, pp. 15-26, TD No. 789 (January 2006).
- F. BUSETTI and S. DI SANZO, *Bootstrap LR tests of stationarity, common trends and cointegration,* Journal of Statistical Computation and Simulation, v. 82, 9, pp. 1343-1355, **TD No. 799 (March 2006).**
- S. NERI and T. ROPELE, *Imperfect information, real-time data and monetary policy in the Euro area,* The Economic Journal, v. 122, 561, pp. 651-674, **TD No. 802 (March 2011).**
- A. ANZUINI and F. FORNARI, *Macroeconomic determinants of carry trade activity*, Review of International Economics, v. 20, 3, pp. 468-488, **TD No. 817 (September 2011).**
- M. AFFINITO, *Do interbank customer relationships exist? And how did they function in the crisis? Learning from Italy*, Journal of Banking and Finance, v. 36, 12, pp. 3163-3184, **TD No. 826 (October 2011).**

- P. GUERRIERI and F. VERGARA CAFFARELLI, Trade Openness and International Fragmentation of Production in the European Union: The New Divide?, Review of International Economics, v. 20, 3, pp. 535-551, TD No. 855 (February 2012).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Network effects of public transposrt infrastructure: evidence on Italian regions, Papers in Regional Science, v. 91, 3, pp. 515-541, TD No. 869 (July 2012).
- A. FILIPPIN and M. PACCAGNELLA, *Family background, self-confidence and economic outcomes,* Economics of Education Review, v. 31, 5, pp. 824-834, **TD No. 875 (July 2012).**

2013

- F. CINGANO and P. PINOTTI, *Politicians at work. The private returns and social costs of political connections*, Journal of the European Economic Association, v. 11, 2, pp. 433-465, **TD No. 709 (May 2009).**
- F. BUSETTI and J. MARCUCCI, *Comparing forecast accuracy: a Monte Carlo investigation*, International Journal of Forecasting, v. 29, 1, pp. 13-27, **TD No. 723 (September 2009).**
- D. DOTTORI, S. I-LING and F. ESTEVAN, *Reshaping the schooling system: The role of immigration*, Journal of Economic Theory, v. 148, 5, pp. 2124-2149, **TD No. 726 (October 2009).**
- A. FINICELLI, P. PAGANO and M. SBRACIA, *Ricardian Selection*, Journal of International Economics, v. 89, 1, pp. 96-109, **TD No. 728 (October 2009).**
- L. MONTEFORTE and G. MORETTI, *Real-time forecasts of inflation: the role of financial variables*, Journal of Forecasting, v. 32, 1, pp. 51-61, **TD No. 767 (July 2010).**
- R. GIORDANO and P. TOMMASINO, *Public-sector efficiency and political culture*, FinanzArchiv, v. 69, 3, pp. 289-316, **TD No. 786 (January 2011).**
- E. GAIOTTI, Credit availablility and investment: lessons from the "Great Recession", European Economic Review, v. 59, pp. 212-227, TD No. 793 (February 2011).
- F. NUCCI and M. RIGGI, *Performance pay and changes in U.S. labor market dynamics*, Journal of Economic Dynamics and Control, v. 37, 12, pp. 2796-2813, **TD No. 800 (March 2011).**
- G. CAPPELLETTI, G. GUAZZAROTTI and P. TOMMASINO, *What determines annuity demand at retirement?*, The Geneva Papers on Risk and Insurance – Issues and Practice, pp. 1-26, **TD No. 805 (April 2011).**
- A. ACCETTURO e L. INFANTE, Skills or Culture? An analysis of the decision to work by immigrant women in Italy, IZA Journal of Migration, v. 2, 2, pp. 1-21, TD No. 815 (July 2011).
- A. DE SOCIO, *Squeezing liquidity in a "lemons market" or asking liquidity "on tap"*, Journal of Banking and Finance, v. 27, 5, pp. 1340-1358, **TD No. 819 (September 2011).**
- M. FRANCESE and R. MARZIA, is there Room for containing healthcare costs? An analysis of regional spending differentials in Italy, The European Journal of Health Economics (DOI 10.1007/s10198-013-0457-4), TD No. 828 (October 2011).
- S. GOMES, P. JACQUINOT, M. MOHR and M. PISANI, Structural reforms and macroeconomic performance in the euro area countries: a model-based assessment, International Finance, v. 16, 1, pp. 23-44, TD No. 830 (October 2011).
- G. BARONE and G. DE BLASIO, *Electoral rules and voter turnout*, International Review of Law and Economics, v. 36, 1, pp. 25-35, **TD No. 833 (November 2011).**
- O. BLANCHARD and M. RIGGI, *Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices*, Journal of the European Economic Association, v. 11, 5, pp. 1032-1052, **TD No. 835 (November 2011).**
- R. CRISTADORO and D. MARCONI, *Household savings in China*, in G. Gomel, D. Marconi, I. Musu, B. Quintieri (eds), The Chinese Economy: Recent Trends and Policy Issues, Springer-Verlag, Berlin, TD No. 838 (November 2011).
- E. GENNARI and G. MESSINA, How sticky are local expenditures in Italy? Assessing the relevance of the flypaper effect through municipal data, International Tax and Public Finance (DOI: 10.1007/s10797-013-9269-9), TD No. 844 (January 2012).
- A. ANZUINI, M. J. LOMBARDI and P. PAGANO, *The impact of monetary policy shocks on commodity prices*, International Journal of Central Banking, v. 9, 3, pp. 119-144, **TD No. 851 (February 2012).**
- R. GAMBACORTA and M. IANNARIO, *Measuring job satisfaction with CUB models*, Labour, v. 27, 2, pp. 198-224, **TD No. 852 (February 2012).**

- G. ASCARI and T. ROPELE, Disinflation effects in a medium-scale new keynesian model: money supply rule versus interest rate rule, European Economic Review, v. 61, pp. 77-100, TD No. 867 (April 2012).
- S. FEDERICO, *Industry dynamics and competition from low-wage countries: evidence on Italy*, Oxford Bulletin of Economics and Statistics (DOI: 10.1111/obes.12023), **TD No. 879 (September 2012).**
- E. BERETTA and S. DEL PRETE, Banking consolidation and bank-firm credit relationships: the role of geographical features and relationship characteristics, Review of Economics and Institutions, v. 4, 3, pp. 1-46, TD No. 901 (February 2013).
- M. ANDINI, G. DE BLASIO, G. DURANTON and W. STRANGE, Marshallian labor market pooling: evidence from Italy, Regional Science and Urban Economics, v. 43, 6, pp.1008-1022, TD No. 922 (July 2013).
- G. SBRANA and A. SILVESTRINI, Forecasting aggregate demand: analytical comparison of top-down and bottom-up approaches in a multivariate exponential smoothing framework, International Journal of Production Economics, v. 146, 1, pp. 185-98, TD No. 929 (September 2013).
- A. FILIPPIN, C. V, FIORIO and E. VIVIANO, *The effect of tax enforcement on tax morale*, European Journal of Political Economy, v. 32, pp. 320-331, **TD No. 937 (October 2013).**

2014

- M. TABOGA, *The riskiness of corporate bonds*, Journal of Money, Credit and Banking, v.46, 4, pp. 693-713, **TD No. 730 (October 2009).**
- G. MICUCCI and P. ROSSI, Il ruolo delle tecnologie di prestito nella ristrutturazione dei debiti delle imprese in crisi, in A. Zazzaro (a cura di), Le banche e il credito alle imprese durante la crisi, Bologna, Il Mulino, TD No. 763 (June 2010).
- M. PERICOLI, *Real term structure and inflation compensation in the euro area*, International Journal of Central Banking, v. 10, 1, pp. 1-42, **TD No. 841 (January 2012).**
- V. DI GACINTO, M. GOMELLINI, G. MICUCCI and M. PAGNINI, *Mapping local productivity advantages in Italy: industrial districts, cities or both?*, Journal of Economic Geography, v. 14, pp. 365–394, TD No. 850 (January 2012).
- A. ACCETTURO, F. MANARESI, S. MOCETTI and E. OLIVIERI, Don't Stand so close to me: the urban impact of immigration, Regional Science and Urban Economics, v. 45, pp. 45-56, TD No. 866 (April 2012).
- M. TABOGA, What is a prime bank? A euribor-OIS spread perspective, International Finance, v. 17, 1, pp. 51-75, **TD No. 895 (January 2013).**
- U. ALBERTAZZI and M. BOTTERO, *Foreign bank lending: evidence from the global financial crisis,* Journal of International Economics, v. 92, 1, pp. 22-35, **TD No. 926 (July 2013).**

FORTHCOMING

- A. MERCATANTI, A likelihood-based analysis for relaxing the exclusion restriction in randomized experiments with imperfect compliance, Australian and New Zealand Journal of Statistics, TD No. 683 (August 2008).
- F. D'AMURI, *Gli effetti della legge 133/2008 sulle assenze per malattia nel settore pubblico*, Rivista di Politica Economica, **TD No. 787 (January 2011).**
- E. COCOZZA and P. PISELLI, Testing for east-west contagion in the European banking sector during the financial crisis, in R. Matoušek; D. Stavárek (eds.), Financial Integration in the European Union, Taylor & Francis, TD No. 790 (February 2011).
- R. BRONZINI and E. IACHINI, Are incentives for R&D effective? Evidence from a regression discontinuity approach, American Economic Journal : Economic Policy, TD No. 791 (February 2011).
- L. GAMBACORTA and P. E. MISTRULLI, *Bank heterogeneity and interest rate setting: what lessons have we learned since Lehman Brothers?*, Journal of Money, Credit and Banking, **TD No. 829 (October 2011).**

- D. FANTINO, A. MORI and D. SCALISE, Collaboration between firms and universities in Italy: the role of a firm's proximity to top-rated departments, Rivista Italiana degli economisti, TD No. 884 (October 2012).
- F. D'AMURI and G. PERI, *Immigration, jobs and employment protection: evidence from Europe before and during the Great Recession,* Journal of the European Economic Association, **TD No. 886** (October 2012).
- L. GAMBACORTA and F. M. SIGNORETTI, *Should monetary policy lean against the wind? An analysis based on a DSGE model with banking*, Journal of Economic Dynamics and Control, **TD No. 921 (July 2013).**
- R. DE BONIS and A. SILVESTRINI, *The Italian financial cycle: 1861-2011*, Cliometrica, **TD No. 936** (October 2013).